Author's response to reviews

Title: Is there an optimal minimally invasive technique for left anterior descending coronary artery bypass?

Authors:

Olivier Jegaden (Olivier.jegaden@chu-lyon.fr)
Fabrice Wautot (fabrice.wautot@chu-lyon.fr)
Thomas Sassard (Thomas.sassard@chu-lyon.fr)
Isabella Szymanik (Isabella.szymanik@chu-lyon.fr)
Abdel Shafy (ashafy@hotmail.com)
Joel Lapeze (joel.lapeze@chu-lyon.fr)
Fadi Farhat (fadi.farhat@chu-lyon.fr)

Version: 2 Date: 30 November 2010

Author's response to reviews:

I modify the article according to a mistake made during upload of the text.